Author's response to reviews

Title: A public health response to the methamphetamine epidemic: The implementation of contingency management to treat methamphetamine dependence

Authors:

Steven Shoptaw (sshoptaw@mednet.ucla.edu)
Jeffrey D. Klausner (Jeff.Klausner@sfdph.org)
Cathy J. Reback (rebackcj@aol.com)
Stephen Tierney (stierney@sfaf.org)
John D Stansell (jstansell@php.ucsf.edu)
Bradley Hare (chare@php.ucsf.edu)
Steven Gibson (sgibson@magnetsf.org)
Michael Siever (msiever@itsa.ucsf.edu)
William D. King (wdking37@yahoo.com)
Uyen Kao (ubui@mednet.ucla.edu)
Jeffrey Dang (jdang@mednet.ucla.edu)

Version: 3 Date: 15 August 2006

Author's response to reviews:

Thank you for the great response in the review. The following details changes to the manuscript submitted.

1. Changed the title to sentence case in the manuscript and in the submission system.
2. Removed degree titles from authors on the title page of the manuscript.
3. Put the country on all author affiliations in the manuscript.
4. Deleted the running head from all pages. Left the page number.
5. Deleted the word count from the title page.
6. Deleted the table count from the title page.
7. Used sentence case for all section headings and titles.
8. All authors read and approved the final manuscript.
9. The website address was included as new reference 10 in the manuscript and references reordered in the text and in the reference section.
10. Provided the full list of authors on the one reference with the et al.
11. I tried to remove the additional file cover letter from the submission page.